Author’s response to reviews

Title: Choledochoduodenal fistula presenting with pneumobilia in a gallbladder cancer patient: A case report

Authors:

Elham Dadzan (elham_d@hotmail.com)
Hossein Akhondi Asl (h68akhond@hotmail.com)

Version: 4 Date: 24 October 2011

Author’s response to reviews:

Hello,

Thanks very much for reviewing our manuscript. We made the following changes based on your input:

1) Extensive discussion about role of ERCP added based on reviewer number 1 comment.
2) The article mentioned by reviewer number 1 cited and added to references.
3) The recommended sentences by reviewer number 1 were added to text.
4) Spelling check was redone. Figure 1 legend mis-spelling corrected.
5) SI units were changed as directed.
6) Separate subheadings for the introduction, case presentation and conclusion were inserted in the abstract.
7) The abbreviations list were incorporated in the body of table rather than a separate entity.
8) Study design was included in the title as mentioned. A presenting with B in C: a case report.
9) No data in the figures can reveal patient's identity.
10) Manuscript was revised and changes are highlighted with yellow colour.
11) This cover letter includes point by point response to the concerns.

Thanks very much again for your time and consideration of this work. Please let me know if further changes are required.

Hossein Akhondi MD